Author’s response to reviews

Title: Acute transient freezing of gait in a patient with posterior reversible encephalopathy syndrome

Authors:

Asuka Nakajima (hot-daro@hotmail.co.jp)
Yuji Ueno (yuji-u@juntendo.ac.jp)
Hideki Shimura (miurashimura@yahoo.co.jp)
Taiki Kambe (tkambe@juntendo.ac.jp)
Kenya Nishioka (kenya.nishioka@gmail.com)
Nobutaka Hattori (nhattori@juntendo.ac.jp)
Takao Urabe (t.urabe@juntendo.ac.jp)

Version: 2 Date: 18 May 2013

Author’s response to reviews:

May 17, 2013
Selim Benbadis
University of South Florida, USA

Dear Professor. Selim Benbadis,

Thank you for your e-mail of May 9, 2013. We were pleased to know your positive evaluation of our manuscript and its potential acceptance for publication in the BMC Neurology, subject to adequate revision and response to the reviewers’ comments. According to the additional comments raised by the reviewers, we modified the manuscript. All changes in the manuscript have been highlighted. All authors have read and approved the manuscript.

We hope this manuscript is satisfactory for publication in “Case Reports in BMC Neurology”.

Very sincerely yours

Yuji Ueno, MD, PhD
Department of Neurology, Juntendo University Urayasu Hospital,
2-1-1 Tomioka, Urayasu, Chiba 279-0021, Japan
Tel: +81-47-353-3111; Fax: +81-47-304-8615
E-mail: yuji-u@juntendo.ac.jp